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        Wed Jul 30 17:06:38 GMT 1997
Year: 97  Doy: 211
Observer: koon
WEATHER COMMENT: Wed Jul 30 17:06:51 GMT 1997
Very rainy, thunderstorms all night, looks like power went out at least
once, wind=20 mph from the SE, temp=43 F.  No observing today.
COMMENT: Wed Jul 30 17:09:04 GMT 1997
Some water leaked in under the door since the new slab is at the same level
as the inside floor, but it was a lot less water than I expected to see,
I mopped it up.  I will continue to look around for other problems to fix.
COMMENT: Wed Jul 30 18:58:55 GMT 1997
Reconnected the lightning ground cable outside in case we have more
thunderstorms, but it looks like they’re gone finally.   Checked on some
things for Charlie and Kim.  Mopped more water near the door in the dome,
restarted the CHIP cooler and reset the OCP clock.  Other than that it
looks like we’re in good shape.
LOW-L COMMENT: Wed Jul 30 19:09:08 GMT 1997
Restarted program so it will write to tape L00468 in drive #0 because the
power outage caused a tape drive change.
COMMENT: Wed Jul 30 20:11:04 GMT 1997
Finished checking things out, heading down now.


